How to Use the Vortex C/C++ Front-end

Jeffrey Dean

Department of Computer Science and Engineering University of Washington Box 352350, Seattle, Washington 98195-2350 cecil@cs.washington.edu

February 6, 1996 5:10 pm

This document describes how to use vortex_cc and vortex_CC, the C and C++ front-ends for Vortex. For the most part, these are plug-in replacements for cc or CC. Here’s how to use them:

vortex_cc -c fib.c -- Compile the C filefib.c into fib.rtl
vortex_CC -c fib.cc -- Compile the C++ file fib.cc into fib.rtl
vortex_cc -c fib.c fib2.c -- Compile the two C filesfib.c and fib2.c into fib.rtl and fib2.rtl

To link a program consisting of a bunch of “object” files, ending in .rtl:

vortex_cc -o fib fib.rtl fib2.rtl--“Link” the two “object” filesfib.rtl and fib2.rtlinto a file called fib.main, which when compiled through Vortex will produce an executable called fib

Here’s a Makefile that is a useful example of a set of rules that take fib1.c and fib2.c and compile them into fib.main, a file which when compiled through Vortex will produce an executable called fib:

V_CC=vortex_cc
V_LINKER=vortex_cc
ROOT_OBJS = fib fib2
OBJS = $(ROOT_OBJS:%=%.o)
RTL_OBJS = $(ROOT_OBJS:%=%.rtl)

%.o: %.c
        ${CC} ${CFLAGS} -c $<

%.rtl: %.c
        ${V_CC} ${CFLAGS} -c $<

default:  fib  fib.main

fib: $(RTL_OBJS)
        ${CC} ${LD_FLAGS} -o fib ${OBJS}

fib.main: $(RTL_OBJS)
        ${V_LINKER} ${LD_FLAGS} -o fib ${RTL_OBJS}